Accelerator Status Report

Run Schedule: Sept. 27, 2010 to December 23, 2010
Oct. 20, 2010
Injector    Energy 61.76 MeV
· The reproducibility of the injector orbits after the Gun HV off/on is still a problem. The experts have not really understood why the GUN behaves like this. The temporary solution is turning off or on the Gun slowly. The Gun control software was modified so that the Gun HV takes about three minutes to ramp down to zero when turning off, and it takes about 20 minutes to ramp up to -130kV when turning on (conditioning mode). During the PSS state changes if we need turn off the Gun HV we need to wait the Gun HV to ramp down to zero before changing the states.
· (10/18) The GUN#2 photocathode was reactivated to boost the QE. The photocathode was damaged during troubleshooting the reproducibility issues of the injector orbits.

· (10/18) ITV0R04 was replaced. 

· (10/11) Injector was set up for three beam operations.
· (10/11) BCM 0L02 was calibrated to FCup#2 up to 100uA.

· (10/11) The Pre-Buncher amplifier malfunctioned and was replaced. The Pre-Buncher setting is different now.

· Orbit reproducibility after the GUN trips is much better now.

Accelerator   Linac Energy 549.00 MeV
· The main machine has been stable.

· The vacuum in zone 1L12 region is not stable, which causes VBV1L12A to close.
· (10/18) ITV2E02 was fixed. The view shows the wires clearly.
· (10/18) EES did some maintenance on the IOTs, which tripped off several times.

· RF zone 2L10 was bypassed twice and EES recovered it. 
· The machine optics is good.

· (10/11) Arc2 box supply tripped several times and the supply was replaced. Afterwards the field maybe a bit off so the machine energy was a bit off after beam was restored. As a result Hall B Tagger was adjusted about 10A to compensate for the energy change. Also power supply (Arc9) and shunt (5S03) caused some down time.
· (10/7) VBV2L17 faulted and would not open due to failed position indicator switch. 
Hall A---Pass 5, DVCS
· The hall has been taking low current (a few uA).

· Y. Roblin tuned Hall A Compton and the hall had very good Compton rates.
· The beam is good.
Hall B ---Pass 5, Primex II
· The hall has been having problems with its detector trigger rates since last Thursday

swing shift. The trigger rates have been too high and sometimes unstable. (10/19) It was found that the hall changed the polarity of the pair spectrometer last Thursday and changed it back later. Also the pair spectrometer tripped off around 0520 on 10/19. After turning on the
pair spectrometer the trigger rates were go for a short period and then they became high and unstable. We do not know the root cause yet. We are monitoring the trigger rates and the Pair Spectrometer current on the Wall. 
· The hall beam is good.

· Do not adjust Hall B Tagger setting. Keeping the same field of the Tagger magnet is 

      important for the Hall.
Hall C--- Pass 1, QWeak 
· (10/20) The hall has been having target problem since this morning. The hall will be down today to address the target issue. 

· (10/19) The hall did some Compton tuning.  

· (10/19) Hall C Moller quads are off due to LCW issues. The hall line optics was adjusted to compensate for the change.

· (10/18) The end stations LCW line was re-configured to provide proper cooling water to Hall C magnets.

· The hall has been taking up to 50uA beam. When sending low current (nA range) to the hall, use attenuator to adjust the current if possible. BPM09BX misbehaves sometimes when using slit to control the current. This could be caused by Hall C laser phase drift.
· Hall C raster X plane sometimes gives zero current. Pushing ‘Send Current’ button will get the current back.
· The beam is good.

· Hall started to beam during the weekend.

